Author’s response to reviews

Title: Healthcare Providers’ Experiences in Supporting Community-Living Older Adults to Manage Multiple Chronic Conditions: A Qualitative Study

Authors:
Jenny Ploeg (ploegj@mcmaster.ca)
Marie-Lee Yous (yousm@mcmaster.ca)
Kimberly Fraser (kimberly.fraser@ualberta.ca)
Sinéad Dufour (sdufour@mcmaster.ca)
Lisa Garland Baird (garlandb@ualberta.ca)
Sharon Kaasalainen (kaasal@mcmaster.ca)
Carrie McAiney (carrie.mcainey@uwaterloo.ca)
Maureen Markle-Reid (mreid@mcmaster.ca)

Version: 2 Date: 31 Oct 2019

Author’s response to reviews:

Dr. Stephen Madamba
Editor, BMC Geriatrics
October 31, 2019

Dear Dr. Madamba,

Thank you for your note of October 29 regarding corrections needed to our manuscript titled Healthcare Providers’ Experiences in Supporting Community-Living Older Adults to Manage Multiple Chronic Conditions: A Qualitative Study. We have addressed the necessary corrections as described below and attached a final version of the manuscript for your consideration,

This paper has not been submitted elsewhere.

With many thanks for your consideration of this work,

Sincerely Yours,
Jenny Ploeg, RN PhD
Professor
Scientific Director, Aging, Community and Health Research Unit
School of Nursing, Faculty of Health Sciences
and Associate Member, Department of Health, Aging and Society
McMaster University
Room HSc3N25C
1280 Main Street West
Hamilton, ON Canada L8S 4K1
Courier Mailing address:
1200 Main Street West, HSC 3N25
Hamilton, Ontario L8N 3Z5
Email: ploegj@mcmaster.ca
Phone: (905) 525-9140 ext 22294
Fax: (905) 524-5199
Website: https://achru.mcmaster.ca/

Editor Comments
1. In the Ethics approval and consent to participate section, please confirm if written informed consent for participation was obtained from all participants.

Response: We have added in this section that written informed consent for participation was obtained from all participants.

2. In the Consent for publication section, please clarify if written informed consent for the publication of quotes was obtained.

Response: We have added in this section that written informed consent for the publication of quotes was obtained.

3. In the Funding section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis and interpretation of data and in writing the manuscript.

Response: We have added in this section that the funding body was not involved in the design of the study and collection, analysis and interpretation of data nor in writing the manuscript.